Author's response to reviews

Title: The influence of ethnic group composition on focus group discussions

Authors:

Nan Greenwood (Nan.Greenwood@sgul.kingston.ac.uk)
Theresa Ellmers (t.ellmers@gmail.com)
Jess Holley (jessholley2108@gmail.com)

Version: 4  Date: 21 August 2014

Author's response to reviews:

The title has been changed as suggested by the editor to: The influence of ethnic group composition on focus group discussions

After looking at the RATs guidelines we have inserted one sentence highlighting that anonymity and confidentiality were assured. page 7 line 17 'Participant anonymity and confidentiality were highlighted.' Otherwise we have adhered to the guidelines.